Author's response to reviews

Title: Study protocol of cost-effectiveness and cost-utility of a biopsychosocial multidisciplinary intervention in the evolution of non-specific sub-acute low back pain in the working population: cluster randomised trial

Authors:

Anna Berenguera (aberenguera@idiapigol.org)
Enriqueta Pujol-Ribera (epujol@idiapigol.org)
Teresa Rodríguez-Blanco (trodriguez@idiapigol.org)
Concepció Violan (cviolan@idiapigol.org)
Marc Casajuana (mcasajuana@idiapigol.org)
Nelleke DeKort (dekortnelleke@gmail.com)
Marta Trapero-Bertran (marta.trapero-bertran@brunel.ac.uk)

Version: 9 Date: 11 August 2011

Author's response to reviews: see over
Dear Editor,

We refer again our manuscript entitled “Study protocol of cost-effectiveness and cost-utility of a biopsychosocial multidisciplinary intervention in the evolution of non-specific sub-acute low back pain in the working population: cluster randomised trial” which we would like to be considered for publication in the BMC Musculoskeletal Disorders.

We have revised our manuscript following the editor’s comments and we have done the major and minor changes on the format of the manuscript.

We look forward to your opinion as to the suitability of new version of the manuscript for inclusion in BMC Musculoskeletal Disorders.

************************

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs) 
The authors has been written exactly as they are in the submission system.

Authors contributions:

It has been added the following sentence: All the authors read and approved the final manuscript.

Minor revisions

Tables- please remove the visible vertical lines from your tables
It has been removed all the vertical lines from the tables

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements 
It has been the possible to re-layout the table 4

Looking forward for your acceptance,

Enriqueta Pujol-Ribera